Author's response to reviews

Title: Clinical characteristics and management of patients with tuberculous pericarditis in Africa in the HIV era

Authors:

Bongani M Mayosi Dr (bmayosi@uctgsh1.uct.ac.za)
Charles Shey Wiysonge Dr (wiysonge@uctgsh1.uct.ac.za)
Mpiko Ntsekhe Dr (mntsekhe@uctgsh1.uct.ac.za)
Jimmy A Volmink Prof (ivolmink@cormack.uct.ac.za)
Freedom Gumedze Mr (gumedze@stats.uct.ac.za)
Gary Maartens Prof (gary@curie.uct.ac.za)
Akinyemi Aje Dr (ajeyemi@yahoo.co.uk)
Baby M Thomas Dr (bmookens@worldonline.co.za)
Kandathil M Thomas Dr (thomas@wildcoast.co.za)
Bongani Thembela Dr (thembelab@xsinet.co.za)
Phindile Mntla Prof (ttisetso@ul.ac.za)
Frans Maritz Dr (maritz@ttctrials.co.za)
Kathleen Ngu Blackett Prof (kathleen@peaslake.abel.co.uk)
Cyrille D Nkouonlack Dr (dukes262001@yahoo.fr)
Vanessa Burch Dr (vanheusden-burch@kingsley.co.za)
Kevin Rebe Dr (krebe@icon.co.za)
Andy Parish Dr (andygp@mweb.co.za)
Karen Sliwa Prof (sliwak@medicine.wits.ac.za)
Brian Vei Dr (brianveyz@ialch.co.za)
Nowshad Alam Dr (nowshadalal@hotmail.com)
Basil G Brown Dr (bgbrown@mweb.co.za)
Trevor Gould Dr (tgould@pgwc.gov.za)
Tim A Visser Dr (tavisser@pgwc.gov.za)
Muki S Shey Mr (shymuk001@mail.uct.ac.za)
Nombulelo Magula Dr (NMagula@tufts-nemc.org)
Patrick J Commerford Prof (picomfrd@uctgsh1.uct.ac.za)

Version: 2 Date: 14 September 2005

Author's response to reviews:

Dear Editor,

I have the honour to re-submit our paper, after making the following changes:

1- All authors are now listed on the front page of the manuscript. We have done away with group authorship.

2- The submitting author is now one of the listed authors on the front page of the manuscript.

Thank you for your time and consideration.

Yours sincerely,

Charles Shey Wiysonge